Author's response to reviews

Title: Diagnostic accuracy of the neurological upper limb examination I: Inter-rater reproducibility of selected findings and patterns.

Authors:

Jorgen R Jepsen (jr@ribeamt.dk)
Lise H Laursen (lisehl@fmm.sdu.dk)
Carl-Goran Hagert (carl-goran.hagert@ort.lu.se)
Svend Kreiner (s.kreiner@biostat.ku.dk)
Anders I Larsen (ail@novozymes.com)

Version: 3 Date: 31 January 2006

Author's response to reviews:

Title, author list and abstract have been reviewed and is now identical in the manuscript and in the submission system. A minor mismatch was due to specific Danish/Swedish letters in authors’ names which could not be recognized in the submission system.

Figure 4a and 4b have been fused into one Figure 4. The text has been revised accordingly.

Vertical text in Table 3 has been changed into horizontal text.

Figure 1 has been cropped.